Author's response to reviews

Title: Two-stage case-control association study of dopamine-related genes and migraine

Authors:

Roser Corominas (rcoromin@ir.vhebron.es)
Marta Ribases (marta.ribases@gmail.com)
Montserrat Camiña (montse@xpereira.com)
Ester Cuenca-León (ecuenca@ir.vhebron.net)
Julio Pardo (jpardof@meditex.es)
Susana Boronat (sboronat@vhebron.net)
María-Jesús Sobrido (mariasobrido@usc.es)
Bru Cormand (bcormand@ub.edu)
Alfons Macaya (amacaya@vhebron.net)

Version: 4 Date: 14 September 2009

Author's response to reviews: see over
Dear Editor,

Next, I list the formatting changes made in our accepted manuscript:

1. Author list: The forename of author Sobrido has been corrected, so that now the author names in the manuscript concur exactly with those in the submission system.

2. References 9, 10 and 34 have been updated with their respective volume and page numbers or “in press” where applicable.

3. As requested, we have deleted any shading from the tables.

4. We have formatted the “Additional Files section” according to the indications provided by the editorial team.

5. Our three additional files are now cited in the text in the way indicated by the editorial team.

6. The keyword list was removed.

Sincerely,

Alfons Macaya MD, PhD
Servei de Neurologia Infantil
Hospital Universitari Vall d'Hebron
08035 Barcelona, Spain